1

2

f o r do c u m e n t at ion on Gnu soft ware ( g cc , g d b ), need to set M A N PA T H

en vironment varia b le to p oint to / u s r/ pr i nc e t on/ m an

( W hat ’s the di �erence b etween a s ystem call and a library routine, e . g.

open vs. fopen ?)

Note : system calls are in section 3 of the man pages

3

$l s

sm all

_ p r o j e c t / large_pro ject/

$ cd small_p roject/

$ lsR

E A

D ME hello

* hello.d hello.output

T

ODO hello. c hello.h make f ile

$ cat README

The"hello

, world" progra m.

makefil e-

builds "hello "

hello.c - the source

hello.h - defines the content of the "hello" string

$ cat hello. c

/* hello.c */#includ e<stdio .h>

#includ e "hello .h"void main

(argc, argv)

int argc;

char *argv[

]; {

printf(" %s

!

\n", HE

LLO_

STR

I

N

G); } /* end of hello.c */

$ cat hello. h

/* hello.h */

#define HELLO_S TRING "Hello, world"

/* end of hello.h */

$ cat TODO

make program say hello in other (human ) langua ges

test program on other comput ers and operatin g systems

$4

$c d l a r g e _ pr o jec t

/

$ ls

R

E A

D ME T

ODO b i n/ src/

$ ls bin

run_

me

* run_me.debug*

$ cd src

$ ls

README driver.c graphics.c include/ main.c make file

$ ls include

macros. h util.h

$5